Author's response to reviews

Title: Peripheral neurolymphomatosis with tracheal asphyxia: a case report and literature review

Authors:
zuofeng liu (zfeng_liu@sina.com)
 tao jiang (suyeku@126.com)
 ni hou (hou_n@sina.com)
yongqian jia (jia_yq@163.com)

Version: 7 Date: 14 July 2015

Author's response to reviews: see over
Author's response to reviews

Title: Peripheral neurolymphomatosis with tracheal asphyxia: a case report and literature review

Authors:
   Zuofeng Liu(zfeng_liu@sina.com)
   Tao Jiang(suyeku@126.com)
   Ni Hou(hou_n@sina.com)
   Yongqian Jia(jia_yq@scu.edu.cn)

Version: 6 Date: 14 July 2015

Author's response to reviews: see over
Reviewer's report
Title: Peripheral neurolymphomatosis with tracheal asphyxia: a case report and literature review
Version: 6 Date: 29 June 2015
Reviewer: Atsushi Fukunaga
Reviewer's report:
Proper comments for my concerns and questions are presented.

Thank you for your affirmation.
Reviewer's report
Title: Peripheral neurolymphomatosis with tracheal asphyxia: a case report and literature review
Version: 6 Date: 8 July 2015
Reviewer: Carlos Kamiya-Matsuoka
Reviewer's report:
Minor Essential Revisions
Clarify "About 60% of patients with NL per reported to have elevated ?? CSF" in line 150
It was revised as “About 60% of patients with NL per reported to have elevated protein level of CSF"

This sentence is redundant "Enhanced MRI is a very useful tool extensively used in urology practice" (152-153)
We revised it as "Enhanced MRI is a very useful tool in neurology practice".

Discretionary Revisions
Protein level of CSF (of around) or fluctuated between 0.54-0.61 g/L, line 87
Done